

Temi di discussione

(Working Papers)

Labor market and financial shocks: a time varying analysis

by Francesco Corsello and Valerio Nispi Landi







Temi di discussione

(Working Papers)

Labor market and financial shocks: a time varying analysis

by Francesco Corsello and Valerio Nispi Landi

Number 1179 - June 2018

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Bassanetti, Marco Casiraghi, Emanuele Ciani, Vincenzo Cuciniello, Nicola Curci, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valentina Michelangeli, Valerio Nispi Landi, Marianna Riggi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

LABOR MARKET AND FINANCIAL SHOCKS: A TIME VARYING ANALYSIS

by Francesco Corsello* and Valerio Nispi Landi*

Abstract

Motivated by the events of the Great Recession, we estimate a time-varying structural VAR model to analyze the effects of a financial shock on the labor market, focusing on the US. Our results indicate that a tightening of financial conditions is highly detrimental to the labor market. Moreover, we show that financial shocks have affected the unemployment rate asymmetrically in the last three decades, an implication that a standard VAR cannot capture: while negative financial shocks have been responsible for increases in unemployment, our model does not find significant contributions of financial shocks during periods of expansion. The source of this asymmetry is the time-varying standard deviation of the identified shock, which is higher in times of financial distress; on the other hand, we find the transmission mechanism is almost constant over time.

JEL Classification: C32, E24, E44. **Keywords**: VAR, labor market conditions, financial markets.

Contents

1. Introduction	5
2. Related literature	9
3. Data	
4. Econometric methodology	
5. Results	
6. Robustness	
7. Conclusions	
Bibliography	
Appendix	

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research.

1 Introduction¹

The Great Recession has been particularly detrimental for the US labor market, in terms of both magnitude and persistence of the effects (Figure 1). The unemployment rate reached a peak of about 10% at the end of 2009 and it fully recovered only in 2016; this employment collapse was mainly driven by an impressive layoffs growth in the private sector; in addition, hours worked per employee have heavily fallen down throughout the crisis and in 2017 they are still not back to their pre-crisis level. Labor force participation was also strongly affected and has accelerated the downward trend started in the early 2000s. Finally, not surprisingly, the vacancy rate experienced a significant drop.

Given that the last recession was generated by a large turmoil in the financial sector (Figure 2), studying the linkages between financial and labor markets seems crucial to gain a better comprehension of business cycle fluctuations. Moreover, the last crisis has fostered a large debate about effectiveness of macroprudential policies aiming to maintain financial stability and to dampen financial shock impact: therefore, analyzing the effects of a financial tightening on the real economy, and especially on the labor market, can provide some useful guidance to policy makers.

In the empirical economic literature there are several works analyzing the macroeconomic effects of a financial shock, defined as a tightening of financial conditions (Gilchrist et al., 2009, and Gilchrist and Zakrajšek, 2012 are eminent examples). These works typically use a VAR model with constant standard deviations (CVAR henceforth) to identify the impact of a financial shock, using sign or short-run restrictions. However, the assumption of constant shock's standard deviation seems quite restrictive when the object of interest is a financial shock: financial time series are typically asymmetric and exhibit changes in volatility over time.² Figure 3 shows the time series of the Chicago Fed National Financial Condition Index (NFCI henceforth) and the Gilchrist and Zakrajsek Spread (GZ, from now on). These indicators measure the tightness of financial conditions in the US, with higher values denoting periods of tighter conditions: both measures feature small fluctuations on average (associated with periods of financial calm) and few big and volatile peaks (associated with periods of financial distress); a VAR with stochastic volatility is able to distinguish between periods of high and low volatility: as a result the estimated

¹We are especially grateful to our supervisor Massimiliano Marcellino for his guidance. We also thank Martin Eichenbaum, Nicola Gennaioli, Francesco Paternò, Giorgio Primiceri, Marco Taboga, Antonella Trigari and Francesco Zanetti for very useful suggestions. All remaining errors are ours. The views expressed in this paper are our own and do not necessarily reflect those of the Bank of Italy.

²See Kim et al. (1998) for an overview of stochastic volatility models.

standard deviation of financial shocks is likely to be high in periods of financial turbulence, as the data seem to suggest. Furthermore, the bias resulting from a CVAR estimation is magnified if the sample period includes the recent financial crisis: the estimated impact of a shock in a CVAR tries to summarize the effects of such a shock over time. Given that the Great Recession faced a huge drop in the real activity due to massive distress in financial markets, a CVAR may overestimate the effect of a financial innovation across the sample. In Section 4 we argue that this overestimation is particularly strong for the unemployment rate.

Nevertheless, time-varying volatility may not be enough to correctly capture the effect of a financial shock. Indeed, not only the size of a financial shock can be time-varying, but also its transmission mechanism. As shown, for instance, by Gaiotti (2013), Hubrich and Tetlow (2015) and Prieto et al. (2016), a change in financial conditions has a higher impact on the real economy during periods of financial distress: this can be explained theoretically by the presence of financial constraints that can be binding during recessions, as pointed out by Mendoza (2010), Bianchi (2011) and Guerrieri and Iacoviello (2017). Moreover, several recent papers show that labor market time series feature a significant degree of skewness.³ Accordingly, as highlighted by Ng and Wright (2013), it is now accepted that parameter instability and stochastic volatility are crucial features that must be considered in a macroeconometric framework.

Motivated by all these considerations, we assess the interactions between labor market and financial shocks in a time-varying framework. In particular, the following research questions are tackled: i) What is the effect of a financial tightening on the labor market? ii) How relevant are financial shocks in explaining historical labor market fluctuations? iii) Is the response of labor market time-varying? iv) Is the size of financial shocks changing over time? And v) what are the main transmission mechanisms and how are they captured by VAR models?

The empirical macroeconomics literature focusing on the time-varying effects of financial shocks in the labor market is quite scant and this is particularly surprising, given what happened during the Great Recession. Therefore, following the work of Primiceri (2005), we estimate a time-varying VAR with stochastic volatility (TV-VAR-SV henceforth), over the sample period 1973-2016. We include in the model eight US quarterly time series: five labor-market variables (unemployment rate, participation rate, hours worked per employee, nominal wage and vacancy rate), two macroeconomic variables (inflation

³See, for instance, McKay and Reis (2008), Ferraro (2017) and Pizzinelli and Zanetti (2017).

rate and real GDP) and the National Financial Condition Index (the NFCI henceforth). The NFCI series and its three components (risk, credit and leverage) are published by the Federal Reserve of Chicago at weekly frequency. This index is a dynamic factor built from a balanced panel of 100 mixed-frequency indicators of financial activity. The financial shock that we aim to identify is a tightening of financial conditions, modeled as an unanticipated exogenous increase in the NFCI: as such, this is a credit supply shock, deeply analyzed by theoretical studies.⁴ The shock is identified through a short-run restriction: the labor market and the macroeconomy cannot simultaneously react to innovations in the NFCI, a standard assumption in the structural VAR literature. This restriction is even more reasonable in our framework, since the labor market is by its own nature sluggish to respond to economic shocks.

Our contribution to the literature is threefold. First we show that a tightening of financial conditions depresses the labor market and the GDP while the effect on inflation is positive and borderline significant.⁵ Second, we find that financial shocks have hit the unemployment rate asymmetrically over the last three decades: while negative financial shocks have been responsible for the high unemployment rates in the early 1980s and during the Great Recession, our model does not find relevant contribution of financial shocks throughout expansion periods. We show that a CVAR is not able to capture this asymmetry. Third, we observe strong evidence of time variation in the financial shock volatility, with peaks during economic downturns, consistently with the works of Justiniano and Primiceri (2008), Stock and Watson (2012) and Prieto et al. (2016): accordingly, we find larger negative than positive shocks. Since the systematic transmission mechanism does not display time variation, we can conclude that the asymmetric response of the unemployment rate is entirely due to financial shock's drifting volatility.

The remainder of the paper is organized as follows: in Section 2 we provide a brief review of theoretical and empirical literatures on the transmission of financial shocks to real activity; Section 3 presents the time series used in the analysis; Section 4 introduces the estimation methodology; Section 5 describes the results; in Section 6 some robustness checks are reported; Section 7 concludes.

⁴See, for instance, the DSGE model by Jermann and Quadrini (2012).

 $^{^5\}mathrm{Notice}$ that the DSGE literature is not unanimous about the effects of financial shocks on the inflation rate.



Figure 1: US labor market during the Great Recession. Unemployment rate, participation rate and wage inflation are expressed in percentage points. Hours per employee is the ratio between hours in nonfarm business sector (an index with base year 2010), and the number of employees in that sector. Layoffs in nonfarm business sector are expressed in thousands of workers. Vacancies are measured by the Help-Wanted Index developed by Barnichon (2010). All these variables but the vacancy index come from the BLS database.



Figure 2: US financial markets during the Great Recession. CBS is the Moody's BAA-AAA corporate bond spread expressed in percentage points. S&P is the Standard & Poor stock market index. House prices are measured by the Standard & Poor's Case–Shiller Home Price Index (base year 2000). NFCI is the National Financial Condition Index developed by the Chicago Fed. Source: St. Louis Fed.



Figure 3: EBP spread is the excess bond premium component of the GZ spread built by Gilchrist and Zakrajšek (2012).

2 Related Literature

2.1 Theoretical Literature

The macroeconomic theoretical literature has deeply analyzed the transmission mechanism linking real activity with the financial sector. The seminal paper by Bernanke and Gertler (1989), Bernanke and Gertler (1995) and Kiyotaki and Moore (1997) introduce the concept of financial accelerator, whereby financial frictions amplify the response of the real economy to economic shocks. These studies formed the basis for business cycle models that incorporate the financial accelerator mechanism in a general equilibrium framework.⁶ In particular, Jermann and Quadrini (2012) develop a model in which the financial sector is not only a source of amplification but also a driver of business cycle fluctuations, as witnessed by the Great Recession. In this model, a tightening of financial conditions reduces firms' borrowing ability and forces them to cut hours of work and production. Also in Gertler and Karadi (2011), the main driver of fluctuations originates in the financial sector: a deterioration in the quality of financial assets reduces the net worth of banks. In turn, banks raise the lending cost to remain profitable and avoid deposit withdrawals. The shock that we aim to identify in our model resembles those in Jermann and Quadrini (2012) and Gertler and Karadi (2011): it is a credit supply shock that makes it harder for firms to borrow funds; using the NFCI goes precisely in this direction, given that this indicator captures the tightness of financial conditions.

A recent strand of the macro-financial literature has departed from the assumption made for computational reasons - that financial constraints are always binding: scholars have been starting to set up models characterized by occasionally binding collateral constraints, in order to study the (potentially) non-linear effects of economic shocks.⁷ During recessions, such constraints bind and firms deleverage; on the other hand, in expansions, a loosening of financial conditions does not have a direct impact on collateral constraints, which continue to be slack.

Our time-varying model points out that the early 1980s and the Great Recession were two periods characterized by negative and highly volatile financial shocks. Some recent papers argue that the time-varying volatility of economic shocks is *per se* a source of macroeconomic fluctuations. In a DSGE framework, Christiano et al. (2014) show that risk shocks, i.e. shocks to the idiosyncratic uncertainty in the return of firms' projects, are an important driver of the business cycle: in this model, higher risk increases the probability of firms' default, accordingly banks raise loan rate and this depresses aggregate demand, triggering the financial accelerator. In a model for an emerging economy, Fernández-Villaverde et al. (2011) analyze the effects of an increase in the interest rate's volatility, which emerging countries take as given: a higher volatility triggers a recession,

⁶See also Bernanke et al. (1999) and Christiano et al. (2003).

⁷For example, occasionally binding collateral constraints characterize the model of Mendoza (2010) and Guerrieri and Iacoviello (2017).

since households cut external debt which becomes more riskier and, as a consequence, reduce consumption. In an estimated DSGE model for the US, Justiniano and Primiceri (2008) argue that the Great Moderation was mainly the result of a reduced volatility in the investment-specific shock, i.e. an exogenous change in the technology transforming investment in capital goods: this is interpreted by the authors as a reduced volatility in financial factors.

A common denominator of these papers is the abstraction from unemployment issues: the labor input is modeled as an intensive margin and so there is no room for unemployment. The Great Recession has provided motivation to fill this gap: frictional financial markets have been introduced in models featuring equilibrium unemployment à la Mortensen and Pissarides (1994): one contribution of our work is to qualitatively capture the dynamics highlighted in these models. A not comprehensive list includes Christiano et al. (2011), Petrosky-Nadeau and Wasmer (2013), Petrosky-Nadeau (2013), Iliopulos et al. (2014) and Zanetti (2017). In these works, financial and labor-matching frictions coexist: in particular, firms need external funding to finance the cost of posting vacancies and attracting unemployed workers. During crises, firms have limited access to external credit and thus reduce the number of posted vacancies, so that financial frictions magnify the role played by search and matching frictions in creating unemployment. Notably, Petrosky-Nadeau (2013) examines the impact of a credit shock on the labor market. In this model, a worsening of financial conditions induces firms to reduce the number of vacancies: the probability to find a job for an unemployed falls and the unemployment rate rises; as a consequence, labor market tightness (defined as the vacancy-unemployment ratio) collapses, making easier for firms to fill a vacancy. The obvious by-product of such a negative shock is a drop of aggregate production. On the other hand, Monacelli et al. (2011) propose a different mechanism. The authors argue that external debt allows firms to get a better position when bargaining wage with workers. In such a case, a negative financial shock forces firms to reduce borrowing and places them in a less favorable bargaining position with workers: as a result, firms will create less jobs.

2.2 Empirical Literature

There is a growing empirical literature analyzing the effects of financial shocks. One seminal paper of this literature is Gilchrist and Zakrajšek (2012): they construct the GZ credit spread index, an indicator with a significant predictive power for economic activity. The GZ spread is decomposed in two components: the expected default and the excess bond premium (EBP); the latter is used as financial variable in a macroeconomic VAR: an innovation in the excess bond premium leads to economically and statistically significant declines in the real activity. Fornari and Stracca (2012) estimate a panel VAR for a large set of advanced economies, in order to evaluate the effects of a credit supply shock, identified with sign restrictions. Their findings show that financial shocks exert a relevant influence on the real economy, both in terms of shock's size and variance explained. These results are confirmed by Caldara et al. (2016): they distinguish between financial and uncertainty shocks, finding that the former ones significantly affect economic conditions and are an important source of business cycle fluctuations; moreover, they amplify the negative effects of adverse uncertainty shocks. One of our goals is contributing to this literature, by augmenting the set of endogenous variables typically included in VAR with some labor-market variables.

The previous works all consider linear frameworks, in which estimated parameters are constant over time. Our work goes one step ahead by allowing time variation in parameters and shock standard deviations. Indeed, some papers point out that including time variation in a financial shock analysis yields some important insights. In particular, many studies find that standard deviation of financial shocks is changing over time and it is relatively higher in periods of financial distress. Using Italian firm-level, Gaiotti (2013) shows that quantitative credit constraints have a larger effect during economic slack. Hubrich and Tetlow (2015) estimate a Markow Switching VAR model, finding that output reacts differently to financial shocks in periods of financial distress; moreover, they also find that shock variances were relatively high during the Great Recession and the dot-com bubble. Prieto et al. (2016) estimate a time-varying VAR for the US economy with three financial market variables (house prices, stock prices and credit spread). They find that during financial crisis, contribution of financial shocks in explaining fluctuations in GDP more than doubles, and this is mainly explained by a higher shock volatility. Alessandri and Mumtaz (2017) argue that a threshold VAR which includes financial indicators outperforms a linear VAR in predicting the probability distribution of US industrial production. Silvestrini and Zaghini (2015) study the linkages between financial shocks and the macroeconomy in a time-varying framework. They estimate a time-varying VAR with euro-area variables, finding evidence of time variation in the transmission mechanism of a shock to the Composite Indicator of Systemic Stress, identified with short-run restrictions: notably, this financial shock has a more significant adverse impact during financial crises. In a reduced-form model, Alessandri et al. (2017) find that changes in the EBP have stronger predictive power for crisis than for expansions.

3 Data

The estimation is performed over the sample period 1973Q1-2016Q3, using eight US time series; the observations going from 1973Q1 to 1980Q4 are used as training sample to calibrate prior distributions. The sample starting point is dictated by availability of the NFCI.

In order to capture financial shocks' effects both on the extensive and intensive margin, the vector of endogenous variables comprises the unemployment rate, the log-first difference of the participation rate (defined as labor force over working age population) and the log-first difference of hours of work per capita in the non-farm business sector; in addition, we include the log-first difference of nominal wage, the vacancy index constructed by Barnichon (2010), GDP deflator and real GDP. Our financial measure is the NFCI.⁸

It deserves to spend some words about the NFCI, given the crucial role played in our analysis. The NFCI, published at weekly frequency by the Federal Reserve Bank of Chicago, is a weighted average of a set of financial indicators: the weight of each indicator is estimated by means of the principal component technique. The index captures measures of risk, liquidity and leverage. The risk component summarizes the premium yielded by risky assets and their volatility; the liquidity component provides an index of willingness to borrow and lend at prevailing prices; the leverage component assesses the economy-wide level of financial debt relative to equity. The index and its components are rescaled to have a zero mean and a unitary standard deviation; positive (negative) values denote financial conditions that are tighter (looser) than average. The main advantage of this indicator comes from an overall description of US financial conditions: this is particularly important due to the strong interconnectivity of US financial markets, in which shocks tend to have an impact on aggregate financial conditions rather than on one specific segment.⁹

In Figure 4 we plot the seven non-financial series we use in estimation, excluding training sample observations. The unemployment rate is above 5% in several periods of our sample:¹⁰ it is particularly high during the 1980s and throughout the Great Recession.

⁸We are not the first ones to use the NFCI in a structural VAR: Fink and Schüler (2015) identify a financial shock as an innovation in the NFCI; Metiu et al. (2015) include the NFCI in a VAR in order to capture the transmission mechanism of a financial shock defined as an innovation to EBP.

 $^{^9 \}mathrm{See}$ Brave and Butters (2011) and Brave and Butters (2012) for more details.

 $^{^{10}\}mathrm{According}$ to FED estimates, the long-run normal rate of unemployment ranges from 4.5 to 5.0 percent.

During the recent financial crisis we also observe unusual negative values for price and wage inflation along with a large drop in hours of work, vacancies and GDP. Figure 5 plots the NFCI and its subcomponents. The index displays positive values at the beginning of the sample (in correspondence with the less-developed-countries debt crisis occurred in the early 1980s), and in the late 1980s, in addition to the big spike in 2009; the years between the 1990's and the boom preceding the Great Recession were characterized by relatively looser financial conditions. The three NFCI components exhibit similar patterns: interestingly, the leverage component is much more volatile, while the credit index is more persistent.



Figure 4: Labor and macroeconomic variables. Variables are expressed in percentage points except vacancies.



Figure 5: NFCI and its subcomponents.

4 Econometric Methodology

4.1 TV-VAR-SV and rank reduction

The n variables of interest are modeled through a time-varying coefficient VAR with stochastic volatility, whose reduced form follows the seminal work of Primiceri (2005):

$$\underbrace{y_t}_{n \times 1} = c_t + \sum_{\ell=1}^p \underbrace{B_{\ell,t}}_{n \times n} y_{t-\ell} + u_t, \qquad u_t \stackrel{i}{\sim} \mathcal{N}\left(\mathbf{0}, \underbrace{\Omega_t}_{n \times n}\right).$$

Following the literature, for estimation purposes the model is transformed as:

$$y_t = X'_t \beta_t + A_t^{-1} \Sigma_t \varepsilon_t, \qquad \varepsilon_t \stackrel{iid}{\sim} \mathcal{N}(\mathbf{0}, I_n),$$
$$\underbrace{X'_t}_{n \times k} \equiv I_n \otimes \begin{bmatrix} 1 & y'_{t-1} & y'_{t-2} & \dots & y'_{t-p} \end{bmatrix}$$

where the time-varying coefficients are stacked in a set of k-dimensional¹¹ vectors $(\beta_t)_{t=1}^T$ and the $n \times n$ reduced form covariances $(\Omega_t)_{t=1}^T$ face a triangular reduction, producing the

 $^{^{11}}k = n(np+1).$

lower triangular matrices $(A_t)_{t=1}^T$ and diagonal matrices $(\Sigma_t)_{t=1}^T$:

$$\underbrace{\beta_t}_{k \times 1} \equiv vec \left(\begin{bmatrix} c_t & B_{1,t} & \dots & B_{p,t} \end{bmatrix}' \right)$$
$$\Omega_t = A_t^{-1} \Sigma_t \Sigma_t \left(A_t^{-1} \right)'.$$

Each group of time-varying elements follows a multivariate random walk with Gaussian innovations, whose covariances are parametrized by matrices $\{Q_{\alpha}, Q_{\beta}, Q_{\sigma}\}$. Indeed, defining the *r*-dimensional vectors¹² $(\alpha_t)_{t=1}^T$ and the *n*-dimensional vectors $(\sigma_t)_{t=1}^T$ as the sets of vectors containing, respectively, the off-diagonal elements in $(A_t)_{t=1}^T$ and the diagonal elements in $(\Sigma_t)_{t=1}^T$, we can write the assumed law of motions:

$\beta_t = \beta_{t-1} + \nu_{\beta,t},$	$ u_{eta,t} \stackrel{iid}{\sim} \mathcal{N}\left(\left. 0, \right. Q_{eta} \left. ight) ight.$
$\alpha_t = \alpha_{t-1} + \nu_{\alpha,t},$	$ u_{\alpha,t} \stackrel{iid}{\sim} \mathcal{N}\left(\ 0, \ Q_{\alpha} \ \right) $
$\log \sigma_t = \log \sigma_{t-1} + \nu_{\sigma,t},$	$ u_{\sigma,t} \stackrel{iid}{\sim} \mathcal{N}(0, Q_{\sigma}). $

As stressed in Primiceri (2005), the random walk assumption is not harmful for estimation purposes, especially since the process is operating for a finite number of periods. Moreover, the random walk specification is parsimonious and does not restrict much the elements' dynamics.

However, a rank reduction strategy proposed by de Wind and Gambetti (2014, WG henceforth) is implemented for the time-varying coefficients. The reason is twofold: first, the rank reduction methodology makes possible to reduce the computational burden of non-small TV-VAR-SV models (say, for larger models than the small one estimated in Primiceri, 2005); second, in line with the findings of Cogley and Sargent (2005) and WG, these models typically do not show large time variation of VAR coefficients when estimated with macro-financial variables. Therefore, a smaller number of components are able to capture coefficients time variation, especially considering that the number of VAR coefficients increases very fast with the number of variables and lags, and there is no reason to conceive such a large number of economic forces able to alter the transmission mechanism at every period.

In particular, Q_{β} is assumed to be a reduced-rank matrix with rank $q_{\beta} < k$ and it is $\frac{1^2r = n(n-1)/2}{2}$

eigendecomposed in the following way:

$$Q_{\beta} = \underbrace{\Lambda_{\beta}}_{k \times q_{\beta}} \Lambda_{\beta}' = \underbrace{V_{\beta}}_{k \times q_{\beta}} \underbrace{D_{\beta}}_{q_{\beta} \times q_{\beta}} V_{\beta}', \qquad \Lambda_{\beta} \equiv V_{\beta} D_{\beta}^{1/2}$$

where D_{β} is a diagonal matrix containing the non-zero eigenvalues of Q_{β} and V_{β} is the matrix where the associated eigenvectors are stacked in columns. Following the procedure of WG, the time-varying coefficient law of motion can be transformed (projecting onto the column space of Λ_{β}) in order to reduce the number of time-varying components to q_{β} . WG highlight that this assumption is reasonable with macroeconomic variables, especially when k gets large; in these circumstances the time variation can be spanned using a smaller space without significant changes in the inference. To sum up, the time-varying coefficients are eventually linear combinations of time-varying components with the following associated laws of motion and time-invariant residual:¹³

$$\underbrace{\beta_t}_{k\times 1} = \Lambda_{\beta}\widetilde{\beta}_t + M_{\beta}\beta_0, \qquad \underbrace{\widetilde{\beta}_t}_{q_{\beta}\times 1} = \widetilde{\beta}_{t-1} + \widetilde{\nu}_{\beta,t}, \qquad \underbrace{\widetilde{\nu}_{\beta,t}}_{q_{\beta}\times 1} \stackrel{iid}{\sim} \mathcal{N}\left(\mathbf{0}, I_{q_{\beta}}\right).$$

Inference on the covariance matrices of states' innovations and on the unobservable states (time-varying coefficients, off-diagonal elements and stochastic volatilities) is produced by means of a Gibbs Sampler that draws from conditional posteriors. The Gibbs Sampler steps are listed below.

Gibbs Sampler Steps¹⁴

- 1. Initialize the Gibbs Sampler at some $(\beta_t^0, \alpha_t^0, \sigma_t^0)_{t=0}^T, Q_{\beta}^0, Q_{\alpha}^0, Q_{\sigma}^0, \text{ set } i = 1.$
- 2. Draw a history of time-varying coefficients $(\beta_t^i)_{t=1}^T$:
 - (a) Draw a history of time-varying components $\left(\widetilde{\beta}_{t}^{i}\right)_{t=1}^{T}$;
 - (b) Draw a vector of time invariant component $M_{\beta}\beta_0$ of $(\beta_t^i)_{t=1}^T$.
- 3. Draw a reduced-rank covariance matrix Q^i_{β} .
- 4. Draw a history of off-diagonal elements $(\alpha_t^i)_{t=1}^T$.

¹³The time invariant residual is obtained through the matrix $M_{\beta} \equiv I_m - \Lambda_{\beta} \left(\Lambda'_{\beta}\Lambda_{\beta}\right)^{-1} \Lambda'_{\beta}$. As shown in Corsello (2018), the residual only depends on the initial condition β_0 .

 $^{^{14}}$ See Corsello (2018) for analytical details on each step.

- 5. Draw a full rank covariance matrix Q^i_{α} .
- 6. Draw a history of volatilities $(\sigma_t^i)_{t=1}^T$.
- 7. Draw a full-rank covariance matrix Q^i_{σ} , set i = i + 1 and restart at [2].

Draws from the posterior of unobservable states are obtained by means of the Carter and Kohn (1994) algorithm. Draws from the Singular Inverse Wishart are implemented differently from what proposed by WG, following the approach of Corsello (2018). The stochastic volatilities draws are implemented as in Primiceri (2005), but respecting the correct order described in Del Negro and Primiceri (2015). The $\log \chi_1^2$ distribution is approximated using the mixture of normal components proposed by Omori et al. (2007), which improves upon the previous standard set by Kim et al. (1998).

The prior distribution is set following Primiceri (2005) for $\{Q_{\alpha}, Q_{\sigma}\}$ and for the initial conditions α_0 and σ_0 , while WG is used as benchmark for calibrating the prior distributions of Q_{β} and β_0 . In particular, the training sample used to calibrate prior distributions runs from 1973Q1 to 1980Q4. Dealing with quarterly data, we set p = 2 lags (as in Primiceri, 2005), so having a total of k = 78 coefficients in the VAR. The reduced rank of the time-varying coefficients is chosen to be $q_{\beta} = 40$, which constitutes a good compromise between dimensionality and span time variation.¹⁵

4.2 Identification Strategy

Our identification scheme relies on a short-run restriction: we assume that both the labor market (unemployment, vacancies, participation, hours and wage) and the macroeconomy (GDP and inflation) do not contemporaneously respond to innovations in the NFCI equation. Accordingly, we order NFCI last in the endogenous vector and impose a Cholesky structure to the matrix of contemporaneous relations.

This identification strategy is quite standard in the economic literature dealing with structural shocks identification. Christiano et al. (1999) and Bernanke et al. (2005) are pioneering examples for the identification of monetary shocks using a Cholesky approach. Gilchrist and Zakrajšek (2012), Hubrich and Tetlow (2015), Prieto et al. (2016) and Silvestrini and Zaghini (2015) use short-run restrictions to identify a financial shock.

¹⁵Increasing the rank does not produce any variation on time-varying coefficient inference, showing that the chosen dimension is sufficient to span the measured time-varying coefficient variations over time.

The rationale behind this assumption is that the private sector typically tends to react with a delay to economic shocks, due to real and nominal rigidities, such as, price and wage adjustment costs or habits persistence in consumption. Clearly, we are well aware that this identification strategy entails some limitations and may lead to biased estimates if these restrictions are too strong or if we are omitting some relevant variables in the endogenous vector of observables. Nevertheless, we think that this approach is a good mix between simplicity and plausibility, also considering that our focus is on labor market, which is sluggish by its own nature. In addition, in Section 6 we estimate the model using monthly data, making the short-run assumption more realistic, although still not completely clean, and we show that results are barely affected.

5 Results

5.1 Impulse responses and stochastic volatilities

Following Lopez-Salido and Nelson (2010) and Prieto et al. (2016), we group quarterly observations available in the sample into financial crisis and non-financial crisis periods. In particular, we consider five periods of US financial distress: the Less-Developed-Countries Debt Crisis (1982-1984), the stock market crash of 1987, the Savings and Loan Crisis (1988-1991), the stock market crash of 2001 and the Global Financial Crisis (2008-2009). For these financially distressed periods (and for the non-crisis observations), we compute the averaged response of each variable to a standardized impulse (unitary increase) to the NFCI.¹⁶ Standardization across periods allows to isolate the transmission mechanism from changes in shock volatilities.

All figures report the 16th and 84th percentiles as bounds for the credible region. We do not detect noticeable time variation in the transmission mechanism, between crisis and non-crisis period, since the response to a standardized impulse is almost constant over time. We find that a one-unit increase in the NFCI significantly increases the unemployment rate by about 0.8% within 10 quarters and depresses hours of work per employee (figure 6). Unemployed workers get discouraged and some of them exit the labor force, given the negative and persistent response of the participation rate. Moreover, wage inflation response shows a significant reduction by about 0.4% at the peak, because of lower labor demand, captured by a significant drop in the vacancy rate. The simultaneous

¹⁶Remind that NFCI is constructed to have zero mean and unitary standard deviation.

reduction in employment and vacancy results in a fall in labor market tightness:¹⁷ according to search and matching models, when labor market is less tight, the job finding rate decreases, while the vacancy filling rate rises. The labor market slack drives a reduction in GDP (by more than 1% in levels at the peak). Interestingly, price inflation is slightly positive and borderline significant: this result is consistent with the VAR estimates by Abbate et al. (2016), who, however, use a different identification strategy.¹⁸

As shown in Figure 7, we find evidence of time-varying volatility of the identified financial shock. In particular, the estimated standard deviation is very high in the beginning of the sample (in correspondence to the Less-Developed-Countries Debt Crisis) and during the Global Financial Crisis. On the other hand, consistently with a large strand of the literature, during the Great Moderation the identified financial shock displays a relatively small standard deviation. In particular, Justiniano and Primiceri (2008) report that the key factor to understand the Great Moderation is the reduction in volatility of the investment shock, defined as innovation to the technology that transforms investments in capital goods. They mention that this innovation may be tied with the cost of external financing for firms, which has shown an important reduction in those years, due to massive financial liberalization.

To better assess the role of drifting volatilities, Figure 8 reports the non-standardized responses to a one-standard-deviation impulse. Time variation is driven by the stochastic volatility that determines the size of the impulse.

As stated previously, there is a growing theoretical literature analyzing the linkages between financial frictions and the labor market. Some of these works focus on the effects of a financial shock on labor market variables: our estimated impulse responses are consistent with these models, at least qualitatively. For instance, in Monacelli et al. (2011) a negative credit shock (by one standard deviation) decreases the employment rate by about

¹⁷Labor market tightness is defined as the vacancy/unemployed ratio. The numerator decreases after the shock; the denominator is affected by two countervailing forces: unemployment rate goes up but labor participation is lower. However, given that the reduction in the participation rate is small in magnitude, we conclude that labor market is less tight following a financial shock.

¹⁸Theoretical models are not unanimous on the effects of a financial tightening on inflation; on the one hand, tighter financial conditions reduce households' expenditure via negative wealth effects, generating a reduction in aggregate demand and so a lower price level (this is the mechanism at work in Gertler and Karadi (2011)); on the other hand, if firms borrow in order to buy intermediate inputs, an increasing risk-premium boosts the marginal cost of firms, which respond by raising the price of their products (as in Atta-Mensah and Dib (2008)). On top of that, Gilchrist et al. (2016) build a model in which firms increase prices when an adverse shock weakens their balance sheets, in order to maintain profitability; this paper provides also empirical evidence showing that, during the Great Recession, firms with a limited internal liquidity decided to raise their prices.

0.35% of the steady state value (which is close to one) at the response peak: this value is close to our median estimate for the crisis periods; GDP and wage responses are close to our median values as well. In Petrosky-Nadeau (2013) a two standard deviations negative credit shock raises the unemployment rate by 2% points at the peak, a value higher than our median estimates during crisis periods; however, both shape and persistence of the reaction are very similar.



Figure 6: Time-varying impulse response functions to a standardized shock NFCI shock.



Figure 7: Stochastic volatility of the identified shock.



Figure 8: Time-varying impulse response functions to a (non-standardized shock) NFCI shock

.

5.2 Historical Decomposition: comparing CVAR and TV-VAR-SV

Having observed significant time variation in the identified financial shock volatility, where larger levels of volatility tend to be associated with financial distress periods, we perform a historical decomposition analysis for the variables of interest.

A CVAR assumes constant volatility of the identified shock. When estimated using financial crisis observations, i.e. when large economic reactions follow financial distress, a CVAR framework may overstate the role of financial shocks in explaining unemployment over the sample. The TV-VAR-SV, instead, is able to discriminate between periods of large and small volatility, complying with the asymmetry of observed financial shocks: large and negative or small and positive.

If we estimate a CVAR using the entire sample (Figure 10), we find that the financial shock's role in explaining unemployment rate fluctuations is roughly as important as for other non-identified shocks. This finding is consistent with Caldara et al. (2016) that use a CVAR to estimate the effects of financial and uncertainty shocks on macroeconomic variables.¹⁹ However, when the CVAR estimation sample ends in 2007Q4 (Figure 9), the picture changes: the financial shock contribution is much smaller across the entire sample period. In these experiments we observe that Great Recession observations are driving the importance of financial shocks across the sample.

Figure 11 displays the historical decomposition of unemployment in a TV-VAR-SV framework. The financial shock hits the unemployment rate asymmetrically: contributions are large and negative in periods of severe financial distress like the Great Recession and the 1982-1984 crisis, while there is almost no positive contributions in the 1990s and early 2000s. This result is in contrast with the previous CVAR analysis and provides solid ground for more general time-varying models when performing structural analysis. This asymmetric behavior is explained by the time-varying volatility of the financial shock, which is estimated to be higher in period of financial distress: indeed, we find that negative shocks tend to be larger and less frequent than positive shocks.

Also for the vacancy rate, negative financial shocks are the main responsible for the fall in 1982-1984 crisis and during the Great Recession, while positive financial shocks cause only mild response in other periods (figure 12). This difference between the CVAR and

 $^{^{19}\}mathrm{See}$ figure B-1 in the working paper version of Caldara et al. (2016).

TV-VAR historical decomposition is less evident for hours and GDP, as shown in the Appendix. Hours of work enter the VAR in (log) first differences: while the level of hours worked per employee is not fully back to the pre-crisis level, the growth rate turned to be positive in 2009 (most recent observations are above the pre-crisis mean).

Accordingly, our identified financial shock was less severe on hours worked in terms of persistence: it is the prolonged negative effect on the unemployment rate that generates this sizable differences between the CVAR and TV-VAR historical decomposition. The same reasoning holds for real GDP, that, like hours worked, enters our model in (log) first differences.



Figure 9: CVAR historical decomposition of unemployment, sample 1983Q1-2007Q4.



Figure 10: CVAR historical decomposition of unemployment, sample 1983Q1-2016Q3.



Figure 11: Historical decomposition of unemployment (TV-VAR-SV).



Figure 12: Historical decomposition of vacancies (TV-VAR-SV).

6 Robustness

In this section we perform three experiments to challenge the robustness of our results. First, we repeat the analysis using the credit spread index built by Gilchrist and Zakrajšek (2012). Second, we estimate the model by using monthly data. Third, we estimate the model excluding the Great Recession from the estimation sample. Most of the resulting figures for these robustness exercises are left in the Appendix.²⁰

6.1 Excess Bond Premium

Gilchrist and Zakrajšek (2012) decompose their credit spread index in two components: the first one captures movements in expected defaults, the second one represents the excess bond premium (EBP, henceforth) which measures cyclical changes in the relationship between default risk and credit spreads. In this subsection we verify the robustness of our results by using EBP as financial variable, as in the VAR analysis of Gilchrist and Zakrajšek (2012); since the correlation between EBP and NFCI is not so high (0.40),

²⁰Additional figures are available upon request.

this represents a good test bench for our findings.²¹ According to the CVAR historical decomposition, the 1990s were a period of positive financial shocks contribution. On the other hand, using a time-varying model, this positive contribution almost disappears, as in our benchmark specification.

Financial shock standard deviation is very low and pretty constant until the late 1990s, when it starts to increase, reaching a peak during the Great Recession (figure 15). Impulse responses seem constant over time, when the impulse is standardized (figure 16). Signs and sizes of responses are in line with the benchmark specification, except for the inflation response, which is negative but not statistically different from zero. Given that stochastic volatility is relatively high mainly in 2001 and during the Great Recession, only during these crises impulse responses functions are time varying, after a non-standardized financial shock. Figures 17 and 18 show the impulse response function of unemployment and vacancy in each crisis period.



Figure 13: Historical decomposition of unemployment (EBP specification, CVAR).

²¹An empirical analysis in a time-varying framework using EBP has been conducted by Abbate and Marcellino (2016), who, however, do not consider the labor market.



 $\label{eq:Figure 14: Historical decomposition of unemployment (EBP specification, TV-SV-VAR).$



Figure 15: Stochastic volatility of the identified financial shock (EBP specification).



Figure 16: Time-varying impulse response functions to an EBP standardized shock.



Figure 17: Time-varying impulse response function of unemployment to an EBP non-standardized shock.



Figure 18: Time-varying impulse response function of vacancy to an EBP non-standardized shock.

6.2 Fixed VAR Coefficients and Monthly Data

Our identification strategy is based on a short-run restriction: variables cannot contemporaneously react to an innovation in NFCI equation residuals. This assumption may seem questionable, if applied to quarterly data. In order to address this issue, we estimate the model using monthly data, keeping fixed the VAR coefficients, thus allowing time variation only in shocks' standard deviations. Indeed, when the model is estimated at monthly frequency, the number of coefficients to estimate gets huge. This is because, when the frequency switches from quarterly to monthly, the number of observations is multiplied by three, causing an increase in the number of lags p, as well as an increase of periods in which k = n (np + 1) coefficients have to be estimated. However, since the benchmark specification does not find much time variation in VAR coefficients, a CVAR with stochastic volatility seems a good compromise.

The Gibbs Sampler to estimate a CVAR with stochastic volatility is a modification of the counterpart for the TV-VAR-SV described above, in which the coefficients' draw step is simply a Bayesian regression with informative prior, obtained by stacking the model in the following form:

$$\underbrace{Y}_{nT\times 1} = \underbrace{X}_{nT\times k} \cdot \beta + U, \qquad U \sim \mathcal{MN} \left(\mathbf{0}, \underbrace{V_u}_{n\times n} \right)$$

where V_u is conditional on the history of stochastic volatilities drawn in the previous step. Prior information is incorporated in the posterior as suggested by Gelman et al. (2013).

Some of the series used in the baseline specification are not available at monthly frequency. The new series that we use are the following:

- Average hours of production and nonsupervisory employees in the private sector.
- Average hourly earnings of production and nonsupervisory employees in the private sector.
- The consumption price index.
- The industrial production.

The unemployment rate, the participation rate and the NFCI are available at monthly frequency, while the vacancy index is not and it is dropped from the estimation sample. All the transformations are the same of the baseline specification, except for hours worked that now are not divided for number of employees, being an average measure.

Results from this new specification do not change the picture: positive financial shocks have a mild impact on unemployment fluctuations (figure 20) and this result cannot be captured by a VAR with constant shocks' variance (figure 19). The estimated stochastic volatility has a pattern similar to the correspondent graph of the quarterly baseline specification (figure 21). The impulse response functions of unemployment, hours, GDP and inflation do not qualitatively differ from the baseline specification (figure 22). Now nominal wage inflation grows on impact; nevertheless, given the inflation response, real wage decreases and this is consistent with a fall in labor demand by firms.



Figure 19: Historical decomposition of unemployment (CVAR, monthly specification).



Figure 20: Unemployment historical decomposition (CVAR-SV, monthly specification).



Figure 21: Stochastic volatility of the identified shock (monthly specification).



Figure 22: Impulse response functions (non-standardized, monthly specification).

6.3 Excluding the Great Recession

In order to assess how much the Great Recession is driving our results, we re-estimate the model by ending the estimation sample in 2007Q4. The historical decomposition of unemployment and the plot of stochastic volatility (figures 23 and 24) resemble their full-sample counterparts. Therefore, the only period in which financial shocks contribute to explain unemployment fluctuations was the 1980s. Moreover, the financial shock's volatility, after the initial peak, has maintained low until the end of the (truncated) sample. Impulse response functions are quantitatively and qualitatively similar to the full-sample counterparts (figure 25). Thus, our results do not seem affected by the Great Recession period.



Figure 23: Unemployment historical decomposition (TV-VAR-SV, no Great Recession).



Figure 24: Stochastic volatility of the identified shock (no Great Recession).



Figure 25: Impulse response functions (standardized, no Great Recession).

7 Conclusions

In this paper we estimate a TV-VAR-SV in order to detect the effects of financial shocks on labor market variables, using the NFCI as financial indicator.

Differently from a CVAR framework, all the elements of a TV-VAR-SV are potentially time-varying: our findings report a relevant degree of time variation in the estimated financial shock's standard deviation, which is high in correspondence of periods of financial distress and low during the Great Moderation. However, our model does not find significant time variation in the transmission mechanism, since the estimated autoregressive coefficients are highly stable over time.

We show that positive financial shocks have been almost irrelevant in explaining unemployment fluctuations in the last three decades, while negative shocks have given a remarkable contribution. This result relies on the changing size of financial shocks, which is high during periods of financial turbulence, such as the Great Recession. This feature cannot be captured by a CVAR, which indeed finds both positive and negative contributions of financial shocks on the unemployment rate, in our estimation horizon.

In addition, we show that estimated response signs to an adverse financial impulse, identified with a short-run restriction, are reasonable and in line with the theoretical literature. A financial tightening generates a crisis in the labor market: firms reduce unemployment and per-capita hours worked to face the reduction in borrowing ability; they post less vacancies, so that resulting lower labor demand produces a fall in nominal wages; some unemployed workers get discouraged and the labor participation rate decreases; the response of real GDP is negative as well, due to a drop in labor input and a smaller demand of households, who become poorer. The response of inflation is positive but only borderline significant, reflecting both supply and demand effects at work.

From a policy perspective, our findings point out that macroprudential measures limiting the probability of a financial crisis may yield significant benefits to the economic activity via the labor market channel. Nonetheless, our model can not say anything about the effectiveness of these policies in reducing the standard deviation of financial shocks: thus, including a macroprudential indicator in macro-financial VARs with stochastic volatility seems an additional promising avenue to extend this line of research.

Bibliography

- Abbate, A., Eickmeier, S., and Prieto, E. (2016). Financial Shocks and Inflation Dynamics. CAMA Working Paper, (53/2016).
- Abbate, A. and Marcellino, M. (2016). Macroeconomic Activity and Risk Indicators: An Unstable Relationship. *Mimeo*.
- Alessandri, P., Conti, A. M., and Venditti, F. (2017). The Financial Stability Dark Side of Monetary Policy. *Temi di discussione (Economic working papers)*, Bank of Italy, (1121).
- Alessandri, P. and Mumtaz, H. (2017). Financial Conditions and Density Forecasts for US Output and Inflation. *Review of Economic Dynamics*, 24:66–78.
- Atta-Mensah, J. and Dib, A. (2008). Bank Lending, Credit Shocks, and the Transmission of Canadian Monetary Policy. *International Review of Economics & Finance*, 17(1):159–176.
- Barnichon, R. (2010). Building a Composite Help-Wanted Index. *Economics Letters*, 109(3):175–178.
- Bernanke, B. S., Boivin, J., and Eliasz, P. (2005). Measuring the Effects of Monetary Policy: a Factor Augmented Vector Autoregressive (FAVAR) Approach. *Quarterly Journal of Economics*, 120(1):387–422.
- Bernanke, B. S. and Gertler, M. (1989). Agency Costs, Net Worth, and Business Fluctuation. *American Economic Review*, 79(1):14–31.
- Bernanke, B. S. and Gertler, M. (1995). Inside the Black Box: The Credit Channel of Monetary Policy Transmission. *Journal of Economic Perspectives*, 9(4):27–48.
- Bernanke, B. S., Gertler, M., and Gilchrist, S. (1999). The Financial Accelerator in a Quantitative Business Cycle Framework. *Handbook of Macroeconomics*, 1:1341–1393.
- Bianchi, J. (2011). Overborrowing and Systemic Externalities in the Business Cycle. American Economic Review, 101(7):3400–3426.
- Brave, S. and Butters, R. A. (2011). Monitoring Financial Stability : a Financial Conditions Index Approach. *Economic Perspectives*, 35(1):22–43.
- Brave, S. and Butters, R. A. (2012). Diagnosing the Financial System: Financial Conditions and Financial Stress. *International Journal of Central Banking*, 8(2):191–239.
- Caldara, D., Fuentes-Albero, C., Gilchrist, S., and Zakrajšek, E. (2016). The macroeco-

nomic impact of financial and uncertainty shocks. *European Economic Review*, 88:185–207.

- Carter, C. K. and Kohn, R. (1994). On Gibbs Sampling for State Space Models. Biometrika, 81(3):541–553.
- Christiano, L. J., Eichenbaum, M., and Evans, C. L. (1999). Monetary Policy Shocks: What Have We Learned and to What End? *Handbook of macroeconomics*, 1:65–148.
- Christiano, L. J., Motto, R., and Rostagno, M. (2003). The Great Depression and the Friedman-Schwartz Hypothesis. Journal of Money, Credit, and Banking, 35(6):1119– 1198.
- Christiano, L. J., Motto, R., and Rostagno, M. (2014). Risk shocks. *American Economic Review*, 104(1):27–65.
- Christiano, L. J., Trabandt, M., and Walentin, K. (2011). Introducing Financial Frictions and Unemployment into a Small Open Economy Model. *Journal of Economic Dynamics* and Control, 35(12):1999 – 2041.
- Cogley, T. and Sargent, T. J. (2005). Drifts and Volatilities: Monetary Policies and Outcomes in the Post WWII US. *Review of Economic dynamics*, 8(2):262–302.
- Corsello, F. (2018). Rank Reduction in BVARs with Time Varying Coefficients and Stochastic Volatility: Specification and Estimation. *Bocconi University, mimeo*.
- de Wind, J. and Gambetti, L. (2014). Reduced-Rank Time-Varying Vector Autoregressions. *CPB Discussion Paper*, (270).
- Del Negro, M. and Primiceri, G. E. (2015). Time Varying Structural Vector Autoregressions and Monetary Policy: a Corrigendum. *Review of Economic Studies*, 82(4):1342– 1345.
- Fernández-Villaverde, J., Guerrón-Quintana, P., Rubio-Ramirez, J. F., and Uribe, M. (2011). Risk Matters: The Real Effects of Volatility Shocks. *American Economic Review*, 101(6):2530–61.
- Ferraro, D. (2017). The Asymmetric Cyclical Behavior of the U.S. Labor Market. Arizona State University Working Paper.
- Fink, F. and Schüler, Y. S. (2015). The Transmission of US Systemic Financial Stress: Evidence for Emerging Market Economies. *Journal of International Money and Finance*, 55:6–26.
- Fornari, F. and Stracca, L. (2012). What Does a Financial Shock Do? First International

Evidence. *Economic Policy*, 71(27):407–445.

- Gaiotti, E. (2013). Credit Availability and Investment: Lessons from the Great Recession. European Economic Review, 59:212–227.
- Gelman, A., Carlin, J., Stern, H., Dunson, D., Vehtari, A., and Rubin, D. (2013). Bayesian Data Analysis, Third Edition. CRC Press.
- Gertler, M. and Karadi, P. (2011). A Model of Unconventional Monetary Policy. *Journal* of Monetary Economics, 58(1):17–34.
- Gilchrist, S., Schoenle, R., Sim, J., and Zakrajšek, E. (2016). Inflation Dynamics During the Financial Crisis. *NBER Working Paper*, (No. w22827).
- Gilchrist, S., Yankov, V., and Zakrajšek, E. (2009). Credit Market Shocks and Economic Fluctuations: Evidence from Corporate Bond and Stock Markets. *Journal of Monetary Economics*, 56(4):471–493.
- Gilchrist, S. and Zakrajšek, E. (2012). Credit Spreads and Business Cycle Fluctuations. American Economic Review, 102(4):1692–1720.
- Guerrieri, L. and Iacoviello, M. (2017). Collateral Constraints and Macroeconomic Asymmetries. *Journal of Monetary Economics*, 90:28–49.
- Hubrich, K. and Tetlow, R. J. (2015). Financial stress and economic dynamics: The transmission of crises. *Journal of Monetary Economics*, 70:100–115.
- Iliopulos, E., Langot, F., and Sopraseuth, T. (2014). Welfare Cost of Fluctuations: when Labor Market Search Interacts with Financial Frictions. *Documents de travail du Centre* d'Economie de la Sorbonne 14042.
- Jermann, U. and Quadrini, V. (2012). Macroeconomic Effects of Financial Shocks. American Economic Review, 102(1):238–71.
- Justiniano, A. and Primiceri, G. E. (2008). The Time-Varying Volatility of Macroeconomic Fluctuations. *American Economic Review*, 98(3):604–41.
- Kim, S., Shephard, N., and Chib, S. (1998). Stochastic Volatility: Likelihood Inference and Comparison with ARCH Models. *Review of Economic Studies*, 65(3):361–393.
- Kiyotaki, N. and Moore, J. (1997). Credit Cycles. *Journal of Political Economy*, 105(2):211–248.
- Lopez-Salido, D. and Nelson, E. (2010). Postwar Financial Crises and Economic Recoveries in the United States. *Working Paper, FED Board*.

- McKay, A. and Reis, R. (2008). The Brevity and Violence of Contractions and Expansions. Journal of Monetary Economics, 55(4):738–751.
- Mendoza, E. G. (2010). Sudden Stops, Financial Crises and Leverage. The American Economic Review, 100(5):1941–1966.
- Metiu, N., Hilberg, B., and Grill, M. (2015). Financial Frictions and Global Spillovers. Bundesbank Discussion Paper, 04/2015.
- Monacelli, T., Quadrini, V., and Trigari, A. (2011). Financial Markets and Unemployment. NBER Working Paper, 17389.
- Mortensen, D. T. and Pissarides, C. A. (1994). Job Creation and Job Destruction in the Theory of Unemployment. *The Review of Economic Studies*, 61(3):397–415.
- Ng, S. and Wright, J. H. (2013). Facts and challenges from the great recession for forecasting and macroeconomic modeling. *Journal of Economic Literature*, 51(4):1120–54.
- Omori, Y., Chib, S., Shephard, N., and Nakajima, J. (2007). Stochastic Volatility with Leverage: Fast and Efficient Likelihood Inference. *Journal of Econometrics*, 140(2).
- Petrosky-Nadeau, N. (2013). Credit, Vacancies and Unemployment Fluctuations. *Review* of *Economic Dynamics*, 17(2):191–205.
- Petrosky-Nadeau, N. and Wasmer, E. (2013). The Cyclical Volatility of Labor Markets under Frictional Financial Markets. American Economic Journal: Macroeconomics, 5(1):193–221.
- Pizzinelli, C. and Zanetti, F. (2017). State Dependence in Labor Market Fluctuations: Evidence, Theory and Policy Implications. *Mimeo*.
- Prieto, E., Eickmeier, S., and Marcellino, M. (2016). Time Variation in Macro-Financial Linkages. Journal of Applied Econometrics.
- Primiceri, G. E. (2005). Time Varying Structural Vector Autoregressions. The Review of Economic Studies, 72(3):821–852.
- Silvestrini, A. and Zaghini, A. (2015). Financial Shocks and the Real Economy in a Nonlinear World: From Theory to Estimation. *Journal of Policy Modeling*, 37(6):915– 929.
- Stock, J. H. and Watson, M. W. (2012). Disentangling the Channels of the 2007–09 Recession. Brookings Papers on Economic Activity, 2012(1):81–135.
- Zanetti, F. (2017). Financial Shocks and Labor Market Fluctuations. *Macroeconomic Dynamics*, (forthcoming).

A Appendix



A.1 Baseline Specification: Additional Figures

Figure A.1: CVAR historical decomposition of vacancy rate, sample 1983Q1-2007Q4.



Figure A.2: CVAR historical decomposition of vacancy rate, sample 1983Q1-2016Q3.







Figure A.4: Historical Decomposition of GDP (TV-VAR-SV).







Figure A.6: Historical Decomposition of hours (TV-VAR-SV).

- N.1152 International financial flows and the risk-taking channel, by Pietro Cova and Filippo Natoli (December 2017).
- N.1153 Systemic risk and systemic importance measures during the crisis, by Sergio Masciantonio and Andrea Zaghini (December 2017).
- N. 1154 Capital controls, macroprudential measures and monetary policy interactions in an emerging economy, by Valerio Nispi Landi (December 2017).
- N. 1155 *Optimal monetary policy and fiscal policy interaction in a non-Ricardian economy*, by Massimiliano Rigon and Francesco Zanetti (December 2017).
- N.1156 Secular stagnation, R&D, public investment and monetary policy: a globalmodel perspective, by Pietro Cova, Patrizio Pagano, Alessandro Notarpietro and Massimiliano Pisani (December 2017).
- N.1157 *The CSPP at work: yield heterogeneity and the portfolio rebalancing channel*, by Andrea Zaghini (December 2017).
- N. 1158 Targeting policy-compliers with machine learning: an application to a tax rebate programme in Italy, by Monica Andini, Emanuele Ciani, Guido de Blasio, Alessio D'Ignazio and Viola Salvestrini (December 2017).
- N.1159 Banks' maturity transformation: risk, reward, and policy, by Pierluigi Bologna (December 2017).
- N. 1160 *Pairwise trading in the money market during the European sovereign debt crisis,* by Edoardo Rainone (December 2017).
- N. 1161 Please in my back yard: the private and public benefitsof a new tram line in *Florence*, by Valeriia Budiakivska and Luca Casolaro (January 2018).
- N. 1162 *Real exchange rate misalignments in the euro area*, by by Michael Fidora, Claire Giordano and Martin Schmitz (January 2018).
- N. 1163 What will Brexit mean for the British and euro-area economies? A model-based assessment of trade regimes, by Massimiliano Pisani and Filippo Vergara Caffarelli (January 2018).
- N. 1164 Are lenders using risk-based pricing in the consumer loan market? The effects of the 2008 crisis, by Silvia Magri (January 2018).
- N.1165 Listening to the buzz: social media sentiment and retail depositors' trust by Matteo Accornero and Mirko Moscatelli (February 2018)
- N. 1166 *Banks' holdings of and trading in government bonds*, by Michele Manna and Stefano Nobili (March 2018).
- N. 1167 Firms' and households' investment in Italy: the role of credit constraints and other macro factors, by Claire Giordano, Marco Marinucci and Andrea Silvestrini (March 2018).
- N. 1168 *Credit supply and productivity growth*, by Francesco Manaresi and Nicola Pierri (March 2018).
- N. 1169 *Consumption volatility risk and the inversion of the yield curve*, by Adriana Grasso and Filippo Natoli (March 2018).
- N.1170 *The global component of inflation volatility*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (April 2018).
- N. 1171 *The potential of big housing data: an application to the Italian real-estate market*, by Michele Loberto, Andrea Luciani and Marco Pangallo (April 2018).
- N. 1172 ECB monetary policy and the euro exchange rate, by Martina Cecioni (April 2018).
- N. 1173 *Firms' investments during two crises*, by Antonio De Socio and Enrico Sette (April 2018).
- N. 1174 How can the government spending multiplier be small at the zero lower bound?, by Valerio Ercolani and João Valle e Azevedo (April 2018).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, Local development that money cannot buy: Italy's Contratti di Programma, Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, Stock market efficiency in China: evidence from the split-share reform, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, TD No. 969 (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, TD No. 896 (January 2013).
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, **TD No. 968 (July 2014).**

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, **TD No. 787 (January 2011).**
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).

- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, Does weak contract enforcement affect firm size? Evidence from the neighbour's court, Journal of Economic Geography, v. 17, 6, pp. 1251-1282, TD No. 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, Is a good example the best sermon? Children's imitation of parental reading, Review of Economics of the Household, v. 15, 3, pp 965–993, D No. 958 (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010).**
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

2018

- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, TD No. 1147 (October 2017).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, TD No. 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, **TD No. 1030 (October 2015).**
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, **TD No. 1025 (July 2015).**
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, **TD No. 1054 (March 2016).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, TD No. 1100 (February 2017).

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038 (November 2015).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces,* Regional Science and Urban Economics, **TD No. 1057 (March 2016).**
- BELOTTI F. and G. ILARDI, Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, **TD No. 1147 (October 2017).**
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, **TD No. 1097 (February 2017).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, **TD No. 909 (April 2013).**
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, TD No. 1008 (April 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, **TD No. 1077 (July 2016).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, **TD No. 1063 (April 2016).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, TD No. 871 (July 2012).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).